Author's response to reviews

Title: Parental divorce in childhood does not independently predict maternal depressive symptoms during pregnancy

Authors:

Elviira Porthan (eespor@utu.fi)
Matti Lindberg (mattlin@utu.fi)
Eeva Ekholm (eevekh@utu.fi)
Noora M. Scheinin (nmsche@utu.fi)
Linnea Karlsson (linnea.karlsson@utu.fi)
Hasse Karlsson (hasse.karlsson@utu.fi)
Juho Härkönen (juho.harkonen@eui.eu)

Version: 2 Date: 28 Aug 2020

Author's response to reviews:

REVISION LETTER

Manuscript PRCH-D-20-00491R1
Original title "Parental separation in childhood does not independently predict maternal depressive symptoms during pregnancy"

Dear Editor, Dear reviewer,

Thank you for your comments. We have now made the recommended change.

RESPONSES TO THE COMMENTS FROM THE EDITOR

1. We note there are missing authors in the Authors' Contributions section. Please include the individual contributions of each author.
A: We have now added the contributions of missing authors in the Authors' Contributions section.